Author's response to reviews

**Title:** The informed consent in southern Italy does not adequately inform parents about infant vaccination

**Authors:**
- Francesco Attena (francesco.attena@unina2.it)
- Amanda Valdes Abuadili (amanda.valdes@unina2.it)
- Sara Marino (sara.marino@unina2.it)

**Version:** 4  **Date:** 7 February 2014

**Author's response to reviews:** see over
7 February 2014

Dear Dr. Silvestre,

Re: Manuscript reference No. 2695535761044527

Please find attached the new revised version of our manuscript titled “The informed consent in southern Italy does not adequately inform parents about infant vaccination”, which we would like to resubmit for publication in *BMC Public Health*

The following page contain my responses to each of the comments of the Editor. For clarity, I present the comments of the editor in full (in italics) and respond to each comment point-by-point. We have also cleaned the text from yellow highlighting.

- *new material in background needs to be references*  
  We have added a new reference in the text (page 3) and in “References”.

- *change spelling of received to receive, in the questions for parents*  
  We have corrected (page 5).

- *the new final statement in the Discussion is not supported by the study and should be deleted*  
  We have deleted the statement (page 10).

I hope that the revisions to the manuscript and the accompanying responses are acceptable, and that the manuscript is suitable for publication.

I look forward to hearing from you.

Yours sincerely,

Francesco Attena

*Address:*
Dipartimento di Medicina Sperimentale  
Via Armanni 5  
80138 Naples  
Italy

[francesco.attena@unina2.it](mailto:francesco.attena@unina2.it)